

GOPEN ACCESS

Citation: Brazier E, Tymejczyk O, Wools-Kaloustian K, Jiamsakul A, Torres MTL, Lee JS, et al. (2024) Long-term HIV care outcomes under universal HIV treatment guidelines: A retrospective cohort study in 25 countries. PLoS Med 21(3): e1004367. https://doi.org/10.1371/journal.pmed.1004367

Academic Editor: Marie-Louise Newell, University of Southampton, UNITED KINGDOM

Received: July 28, 2023

Accepted: February 22, 2024

Published: March 18, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1004367

Copyright: © 2024 Brazier et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Complete data for this study cannot be posted in a supplemental file or a public repository because of legal and ethical restrictions. The principles of collaboration of **RESEARCH ARTICLE**

Long-term HIV care outcomes under universal HIV treatment guidelines: A retrospective cohort study in 25 countries

Ellen Brazier^{1,2}*, Olga Tymejczyk¹, Kara Wools-Kaloustian³, Awachana Jiamsakul⁴, Marco Tulio Luque Torres⁵, Jennifer S. Lee⁶, Lisa Abuogi⁷, Vohith Khol⁸, Fernando Mejía Cordero⁹, Keri N. Althoff⁶, Matthew G. Law¹⁰, Denis Nash^{1,2}, on behalf of the International epidemiology Databases to Evaluate AIDS (leDEA)¹

 City University of New York, Institute for Implementation Science in Population Health (ISPH), New York, New York, United States of America, 2 City University of New York, Graduate School of Public Health and Health Policy, New York, New York, United States of America, 3 Department of Medicine, Indiana University School of Medicine, Indianapolis, Indiana, United States of America, 4 The Kirby Institute, University of New South Wales, Sydney, Australia, 5 Department of Pediatrics, Instituto Hondureño de Seguridad Social and Hospital Escuela Universitario, Tegucigalpa, Honduras, 6 Department of Epidemiology, Johns Hopkins Bloomberg School of Public Health, Baltimore, Maryland, United States of America, 7 Department of Pediatrics, University of Colorado School of Medicine and Children's Hospital Colorado, Aurora, Colorado, United States of America, 8 National Center for HIV/AIDS, Dermatology and STDs, Phnom Penh, Cambodia, 9 Instituto de Medicina Tropical Alexander von Humboldt, Universidad Peruana Cayetano Heredia, Lima, Peru, 10 The Kirby Institute, University of New South Wales, Sydney, Australia

¶ Membership of IeDEA is provided in <u>S1 Text</u>. * ellen.brazier@sph.cuny.edu

Abstract

Background

While national adoption of universal HIV treatment guidelines has led to improved, timely uptake of antiretroviral therapy (ART), longer-term care outcomes are understudied. There is little data from real-world service delivery settings on patient attrition, viral load (VL) monitoring, and viral suppression (VS) at 24 and 36 months after HIV treatment initiation.

Methods and findings

For this retrospective cohort analysis, we used observational data from 25 countries in the International epidemiology Databases to Evaluate AIDS (IeDEA) consortium's Asia-Pacific, Central Africa, East Africa, Central/South America, and North America regions for patients who were ART naïve and aged \geq 15 years at care enrollment between 24 months before and 12 months after national adoption of universal treatment guidelines, occurring 2012 to 2018. We estimated crude cumulative incidence of loss-to-clinic (CI-LTC) at 12, 24, and 36 months after enrollment among patients enrolling in care before and after guideline adoption using competing risks regression. Guideline change–associated hazard ratios of LTC at each time point after enrollment were estimated via cause-specific Cox proportional hazards regression models. Modified Poisson regression was used to estimate relative risks of retention, VL monitoring, and VS at 12, 24, and 36 months after ART initiation. There were 66,963 patients enrolling in HIV care at 109 clinics with \geq 12 months of follow-up time after

leDEA and the regulatory requirements of the different IRBs of our participating sites (sometimes representing national IRBs of ministries of health) require the submission of a project concept proposal and approval by the IeDEA Executive Committee. To request data, please review IeDEA guidance available at: https://www.iedea.org/ resources/multiregional-research-sops-templates/ and contact the Executive Committee (https://www. iedea.org/working-groups/executive-committee/). Signing of a data sharing agreement may also be required.

Funding: The International Epidemiology Databases to Evaluate AIDS (IeDEA) is supported by the U.S. National Institutes of Health's National Institute of Allerov and Infectious Diseases (https:// www.niaid.nih.gov), the Eunice Kennedy Shriver National Institute of Child Health and Human Development (https://www.nichd.nih.gov), the National Cancer Institute (https://www.cancer.gov), the National Institute of Mental Health (https:// www.nimh.nih.gov), the National Institute on Drug Abuse (https://nida.nih.gov), the National Heart, Lung, and Blood Institute (https://www.nhlbi.nih. gov), the National Institute on Alcohol Abuse and Alcoholism (https://www.niaaa.nih.gov), the National Institute of Diabetes and Digestive and Kidney Diseases (https://www.niddk.nih.gov). the Fogarty International Center (https://www.fic.nih. gov), and the National Library of Medicine (https:// www.nlm.nih.gov): Asia-Pacific, U01AI069907 (AJ, MGL and VK); Caribbean, Central and South America network for HIV epidemiology (CCASAnet), U01AI069923 (MTLT and FMC): Central Africa, U01AI096299 (EB, DN and OT); East Africa, U01AI069911 (KWK and LA); NA-ACCORD, U01AI069918 (KNA and JSL). Informatics resources are supported by the Harmonist project, R24Al124872. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: KNA receives royalties from Coursera. She was a consultant to the All of Us Research Program. She serves on the scientific advisory board for TrioHealth Inc. DN received consulting fees from Abbvie and Gilead and is the PI of a research grant from Pfizer to his institution'. All other authors have declared that no competing interests exist.

Abbreviations: aHR, adjusted hazard ratio; aRR, adjusted relative risk; ART, antiretroviral therapy; CI-LTC, cumulative incidence of loss-to-clinic; COVID-19, Coronavirus Disease 2019; GEE, generalized estimating equation; IeDEA, International epidemiology Databases to Evaluate AIDS; IQR, interquartile range; NNRTI, nonenrollment (46,484 [69.4%] enrolling before guideline adoption and 20,479 [30.6%] enrolling afterwards). More than half (54.9%) were females, and median age was 34 years (interguartile range [IQR]: 27 to 43). Mean follow-up time was 51 months (standard deviation: 17 months; range: 12, 110 months). Among patients enrolling before guideline adoption, crude CI-LTC was 23.8% (95% confidence interval [95% CI] 23.4, 24.2) at 12 months, 31.0% (95% CI [30.6, 31.5]) at 24 months, and 37.2% (95% [CI 36.8, 37.7]) at 36 months after enrollment. Adjusting for sex, age group, enrollment CD4, clinic location and type, and country income level, enrolling in care and initiating ART after guideline adoption was associated with increased hazard of LTC at 12 months (adjusted hazard ratio [aHR] 1.25 [95% CI 1.08, 1.44]; p = 0.003); 24 months (aHR 1.38 [95% CI 1.19, 1.59]; p < .001); and 36 months (aHR 1.34 [95% Cl 1.18, 1.53], p < .001) compared with enrollment before guideline adoption, with no before-after differences among patients with no record of ART initiation by end of follow-up. Among patients retained after ART initiation, VL monitoring was low, with marginal improvements associated with guideline adoption only at 12 months after ART initiation. Among those with VL monitoring, VS was high at each time point among patients enrolling before guideline adoption (86.0% to 88.8%) and afterwards (86.2% to 90.3%), with no substantive difference associated with guideline adoption. Study limitations include lags in and potential underascertainment of care outcomes in real-world service delivery data and potential lack of generalizability beyond IeDEA sites and regions included in this analysis.

Conclusions

In this study, adoption of universal HIV treatment guidelines was associated with lower retention after ART initiation out to 36 months of follow-up, with little change in VL monitoring or VS among retained patients. Monitoring long-term HIV care outcomes remains critical to identify and address causes of attrition and gaps in HIV care quality.

Author summary

Why was this study done?

- Although universal HIV treatment recommendations have been adopted in national HIV treatment guidelines, longer-term HIV care outcomes under such guidelines are poorly documented and largely limited to single-country studies with short follow-up times.
- No multicountry studies using real-world service delivery data have examined longterm HIV care outcomes associated under universal HIV treatment guidelines.

What did the researchers do and find?

• With data on 66,963 patients enrolling in HIV care at 109 clinics participating in the International epidemiology Databases to Evaluate AIDS (IeDEA) research consortium across 25 countries where universal HIV treatment guidelines were adopted, we

nucleoside reverse transcriptase inhibitor; PLWH, people living with HIV; VL, viral load; VS, viral suppression; WHO, World Health Organization; 95% Cl, 95% confidence interval.

estimated the hazard ratios of loss-to-clinic (LTC) at 12, 24, and 36 months after enrollment, comparing those enrolling in HIV care after guideline adoption to those enrolling before guideline adoption.

- Among 57,615 patients with documented initiation of antiretroviral therapy (ART), we also estimated the relative risks of clinic retention, viral load (VL) monitoring, and viral suppression (VS) at 12, 24, and 36 months after ART initiation, comparing those enrolling after versus before national adoption of universal treatment guidelines.
- Compared with patients enrolling in HIV care and initiating HIV treatment before national adoption of universal treatment guidelines, those enrolling and initiating treatment after guideline adoption had higher risk of being LTC at 12 months, 24 months, and 36 months after enrollment.
- Among patients retained in care after ART initiation, those enrolling in HIV care after the adoption of universal HIV treatment guidelines were more likely to have VL monitoring at 12 months after ART initiation and less likely at 36 months, with no difference at 24 months.
- VS was high at each time point among patients enrolling before and after the adoption of universal HIV treatment guidelines, with no substantive change associated with guideline adoption.

What do these findings mean?

- Our results raise concerns about long-term retention of patients after ART initiation, as well as the capacity of HIV programs to provide essential aspects of HIV care, including annual VL monitoring for timely identification of adherence problems and treatment failure.
- Our findings that patient retention in care at the clinic where ART was initiated decreased after the adoption of universal HIV treatment guidelines and that there has been no improvement in annual VL monitoring among patients retained in care should motivate efforts to identify and address factors associated with attrition among patients enrolling in HIV care, as well as barriers to routine VL testing in the era of universal treatment of all people living with HIV.
- Study limitations include potential underascertainment of patient outcomes in realworld service delivery data, lags in the availability of real-world service delivery data, and the nonrepresentativeness of the clinics and countries reflected in IeDEA datasets available for analysis.

Introduction

The World Health Organization (WHO)'s 2015 recommendation for universal treatment for all people living with HIV (PLWH) [1]—known as "Treat-All"—eliminated an important barrier to initiation of antiretroviral therapy (ART) in many settings [2]. While a few high-income countries had universal HIV treatment guidelines in place prior to WHO's 2015 recommendation, most countries around the globe adopted expanded treatment guidelines subsequently,

with an estimated 70% of low- and middle-income countries adopting universal HIV treatment guidelines by the end of 2017 [3].

Observational studies have shown that national adoption of universal treatment guidelines has led to greater uptake and more rapid initiation of ART across diverse country settings [4,5]. Improved treatment uptake and more timely initiation of ART are promising for the reduction of HIV morbidity and mortality in patients, as well as preventing onward transmission of the virus (i.e., treatment as prevention) [6–8]. However, longer-term HIV care outcomes, such as retention in care and timely and sustained viral suppression (VS), under universal treatment guidelines are underresearched and largely limited to small single-country studies with short follow-up times of 6 to 12 months [9–14]. While a community cluster-randomized controlled trial in Uganda and step-wedged randomized trial in Eswatini have reported higher 12-month retention and combined 12-month retention/VS rates among patients initiating treatment under universal treatment guidelines, compared with standard initiation practices [15,16], several observational studies have reported no improvement in care retention—or lower retention—among patients initiating treatment in the era of universal treatment [10–13].

Using data from the International epidemiology Databases to Evaluate AIDS (IeDEA) research consortium, we aimed to estimate loss-to-clinic (LTC) at 12, 24, and 36 months after enrollment in HIV care, comparing those enrolling before and after country-level adoption of universal HIV treatment guidelines. Additionally, we aimed to estimate clinic retention, viral load (VL) testing, and VS at 12, 24, and 36 months after ART initiation.

Methods

Data sources

The IeDEA consortium pools observational clinical data on more than 2 million PLWH ever enrolling in HIV care at approximately 400 care and treatment sites in 44 countries [17]. Our study population was drawn from IeDEA's cohorts in the Asia-Pacific, Central Africa, East Africa, the Caribbean, Central and South America, and North America regions, which agreed to the use of their data for this retrospective cohort study, based on a concept proposal approved by IeDEA's executive committee. Deidentified patient data from participating IeDEA cohorts were standardized in accordance with IeDEA data definitions [18]. The research was approved by the City University of New York (CUNY) University Institutional Review Board (#2018–0809).

For each country in participating regional cohorts, we identified the date universal ART eligibility was extended to all adult patients, based on policy documents, literature, and inputs from in-country experts, as described elsewhere [19]. Patients were eligible if they were at least 15 years of age and ART naïve at the time of enrollment in HIV care at an IeDEA site, enrolled in care in the 24 months immediately before national adoption of universal treatment guidelines or in the first 12 months thereafter, and enrolled in care at least 12 months before database closure (i.e., submission of data to IeDEA's regional data centers for processing in accordance with IeDEA's data exchange standards [18]). For cohorts that deidentify patient data by shifting patient encounter dates by <30 days, we excluded all patients enrolling in care within +/- 30 days of the date of guideline adoption. Cohorts where all patient enrollment dates are shifted to a midyear date (i.e., July 1) were excluded, along with clinics where no patients had any records of VL monitoring. Additionally, we excluded patients with missing data for sex or age at enrollment in HIV care, and missing date of death if recorded as deceased. The years of data used in this study ranged from 2010 to 2021.



Fig 1. Study populations and windows for primary outcome ascertainment.

https://doi.org/10.1371/journal.pmed.1004367.g001

Exposure

The exposure of interest was enrollment in HIV care before or after the official date of national adoption of universal HIV treatment guidelines, which, depending on the country, occurred between 2012 and 2018.

Outcomes

Among all patients who were ART naïve, who enrolled in HIV care in the 24 months before or 12 months after national adoption of universal HIV treatment guidelines, and who had sufficient follow-up time between enrollment and database closure, our primary outcome of interest was LTC by 12, 24, and 36 months after enrollment (Fig 1). Across all regional cohorts, LTC was defined as no evidence of contact (e.g., visits, laboratory testing, or medication pick-ups) with the clinic of enrollment for at least 12 months prior to database closure [20]. The date of LTC was set at 90 days after the last clinic contact, with patients considered LTC if their date of LTC was before censoring at 12, 24, and 36 months, respectively. Patients not documented as having died or transferred and not classified as LTC by the censoring date were considered retained in care at the clinic.

Among the subset of patients with evidence of ART initiation and sufficient follow-up time between ART initiation and database closure, we examined clinic retention at 12, 24, and 36 months after ART initiation, with retention defined as no documentation of death or transfer to another site of care, and not LTC (as defined above). Among patients on ART and retained in care, we examined VL monitoring, defined as any VL test at 12, 24, and 36 months (+/–3 months) after ART initiation, and among patients with VL monitoring at these time points, we examined VS, defined as VL <1,000 copies/mL. ART, antiretroviral therapy; LTC, loss-toclinic; VL, viral load; VS, viral suppression.

Covariates

Patient-level characteristics included sex (male or female); age at enrollment in HIV care (categorized as 15 to 19 years, 20 to 24 years, 25 to 34 years, and >34 years; CD4 count within 90 days (+/–) of enrollment and no more than 30 days after ART initiation (categorized as: \leq 200 cells/µL; 201 to 350 cells/µL; 351 to 500 cells/µL; >500 cells/µL; or unknown/missing); ART initiation was defined as the start of a combination antiretroviral treatment regimen before censoring at 12, 24, and 36 months after enrollment, and initial ART regimens were categorized as non-nucleoside reverse transcriptase inhibitor (NNRTI)-based regimens, protease inhibitor– based regimens, integrase inhibitor–based regimens, other/unknown regimens, or none). Clinic-level characteristics included location (rural/mostly rural versus urban/mostly urban); facility type (i.e., health center, district hospital, regional/university referral hospital, or other); and country income level in 2018 (low, lower-middle, upper-middle, and high income) as reflected in World Bank databases [21].

Statistical analysis

We used descriptive statistics to compare the characteristics of patients enrolling before and after adoption of universal HIV treatment guidelines, along with LTC at each time point after enrollment, and retention, VL monitoring, and VS after ART initiation. We also described the number and proportion of patients recorded as having transferred or died by each time point after enrollment.

We estimated the crude cumulative incidence (i.e., risk) of LTC at 12, 24, and 36 months after enrollment, stratified by the timing of enrollment relative to the date of guideline adoption and ART initiation status by the censoring time point (i.e., on ART or not by 12, 24, and 36 months after enrollment) via competing risks regression using the Aalen–Johansen estimator [22]. Multivariable cause-specific Cox proportional hazards regression [23] was used to estimate the association between universal treatment guideline adoption and LTC at 12, 24, and 36 months after enrollment, adjusting for the above covariates, and the clustering of patients within clinics was accounted for when fitting the Cox proportional hazards models through the use of a robust sandwich estimator for the covariance matrix. We considered death as a competing risk for LTC, with transfer treated as a censoring variable. We tested for statistical interactions between enrollment period (before versus after guideline adoption) and ART initiation status by censoring time points and stratified results by ART initiation status (i.e., on ART or not on ART).

Among the subset of patients with evidence of ART initiation, we estimated relative risks of the binary outcomes of clinic retention, VL monitoring, and VS at 12, 24, and 36 months after ART initiation via modified Poisson regression models, comparing those enrolling in care after versus before universal HIV treatment adoption. Multivariable models were adjusted for patient and clinic characteristics (sex, age group, enrollment CD4, initial regimen type, clinic location, facility type, and country income level), using generalized estimating equations (GEEs) to account for clustering within clinics.

In a sensitivity analysis, we compared relative hazards of LTC at 12 months after enrollment and relative risks of care retention, VL monitoring, and VS at 12 months after ART initiation among patients enrolling in HIV care in the 12 months after adoption of universal HIV treatment guidelines versus the 13 to 24 months before guideline adoption (i.e., excluding patients enrolling in care during the 12 months immediately before guideline adoption whose outcome ascertainment window was entirely in the period after guideline adoption) (Fig 2).

An initial concept proposal for this analysis (S1 Concept proposal) outlined exposures, types of outcomes of interest, participating IeDEA cohorts, and general analytic approach and was approved by the IeDEA Executive Committee in August 2018. The use of competing risks and cause-specific hazards regression to examine LTC was not prespecified, with these methods introduced a priori to avoid potential bias that might result from failure to account for competing events in outcome estimation. In a second sensitivity analysis, performed in response to reviewer feedback, we excluded countries where universal treatment guidelines were adopted in 2017 and 2018, to assess whether our estimates of HIV care outcomes at 24 and 36 months could have been affected by the Coronavirus Disease 2019 (COVID-19) pandemic in countries with late adoption of universal HIV treatment guidelines.

All statistical analyses were performed using SAS 9.4 (SAS Institute, Cary, NC). A *p*-value less than 0.05 was considered statistically significant. This study is reported as per the



Fig 2. Sensitivity analysis: Study populations and windows for outcome ascertainment.

https://doi.org/10.1371/journal.pmed.1004367.g002

Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline (<u>S1 Checklist</u>).

Results

Of 132,776 patients in the initial data submission from participating IeDEA regions, 54,961 did not meet eligibility criteria related to age, ART-naïve status, or timing of enrollment, and 10,689 were excluded because they were in care at clinics with no VL testing or in cohorts where patient data are anonymized by shifting dates of care (Fig 3). Additionally, 163 patients were excluded because of missing data related to age, sex, or death date. After these exclusions, there were 66,963 patients who were ART naïve and aged \geq 15 years at enrollment in HIV care at 109 clinics in 25 countries (Argentina, Australia, Brazil, Burundi, Cambodia, Cameroon, Canada, Chile, Congo, Democratic Republic of Congo, Haiti, Honduras, Hong Kong, Japan, Kenya, Malaysia, Mexico, Peru, Rwanda, South Korea, Tanzania, Thailand, Uganda, United States, and Vietnam). The sample included 46,484 (69.4%) who enrolled in the 24 months prior to national adoption of universal HIV treatment guidelines and 20,479 (30.6%) who enrolled in the 12 months after guideline adoption (Table 1). Mean follow-up time was 51 months (standard deviation: 17 months; range: 12, 110 months).

Among the full sample of patients with at least 12 months of follow-up time between enrollment and database closure, 54.9% were female, and the median age was 34 years (interquartile range [IQR]: 27, 43), with little difference by period of enrollment (before vs. after adoption of universal treatment guidelines). A smaller proportion of patients enrolling in care after adoption of universal treatment guidelines had CD4 testing results recorded at enrollment (33.8% after vs. 47.4% before; p < 0.001); however, among those with any enrollment CD4 test, median CD4 counts were clinically similar among those enrolling before and after adoption of universal treatment guidelines (302 cells/µL [IQR: 136, 492] vs. 315 cells/µL [IQR: 138, 518]). The proportion of patients who had initiated ART by 12 months after enrollment was higher among those enrolling in the year after guideline adoption, compared with those enrolling before (87.6% vs. 78.4%; p < 0.001), and among those initiating ART within 12 months of enrollment, the median time from enrollment to treatment initiation decreased from 14 days (IQR: 0, 44) to 0 days (IQR: 0, 14).

The majority of patients (78.8%) were in care at clinics in urban/mostly urban settings, and almost half (45.3%) were at tertiary hospitals, with 24.4% at health centers and 24.7% at district hospitals. Most patients (83.2%) were from low- and lower-middle-income countries and





https://doi.org/10.1371/journal.pmed.1004367.g003

countries that adopted universal HIV treatment guidelines in 2016 (75.4%), with small minorities of patients in countries with earlier (2012 to 2015) or later adoption of such guidelines (2017 to 2018).

Of patients enrolling before national adoption of universal HIV treatment guidelines, 45,098 (97.0%) had at least 24 months of follow-up time before database closure, and 42,859 (92.2%) had at least 36 months of follow-up time. Among patients enrolling after guideline adoption, 19,012 (92.8%) had at least 24 months of follow-up time and 11,392 (55.6%) had at least 36 months of follow-up time. The distribution of patient and clinic characteristics among those with 24 and 36 months of follow-up time after enrollment was similar to those of patients

Patient characteristics	N (%)	Before guideline adoption	After guideline adoption	<i>p</i> -Value ^a
All patients	66,963	46,484 (69.4)	20,479 (30.6)	
Sex				
Male	30,233 (45.1)	20,872 (44.9)	9,361 (45.7)	0.053
Female	36,730 (54.9)	25,612 (55.1)	11,118 (54.3)	
Age (in years)				
Median age (IQR)	34 (27, 43)	34 (27, 43)	34 (27, 43)	0.006 ^b
15–19 years	2,939 (4.4)	2,085 (4.5)	854 (4.2)	0.046
20-24 years	9,203 (13.7)	6,469 (13.9)	2,734 (13.4)	
25–34 years	21,729 (32.4)	15,050 (32.4)	6,679 (32.6)	
>34 years	33,092 (49.4)	22,880 (49.2)	10,212 (49.9)	
CD4 count at enrollment				
No CD4 count at enrollment	38,009 (56.8)	24,455 (52.6)	13,554 (66.2)	< 0.001
Any CD4 count at enrollment	28,954 (43.2)	22,029 (47.4)	6,925 (33.8)	
Median CD4 count (IQR)	304 (136, 497)	302 (136, 492)	315 (138, 518)	< 0.001 ^b
<200 cells/µl	9,972 (34.4)	7,637 (34.7)	2,335 (33.7)	< 0.001
200–349 cells/µl	6,492 (22.4)	5,005 (22.7)	1,487 (21.5)	
350–499 cells/µl	5,359 (18.5)	4,104 (18.6)	1,255 (18.1)	
> = 500 cells/µl	7,131 (24.6)	5,283 (24.0)	1,848 (26.7)	
Initiation of ART by censoring endpoint (12 months)				
Not on ART	12,575 (18.8)	10,041 (21.6)	2,534 (12.4)	< 0.001
On ART	54,388 (81.2)	36,443 (78.4)	17,945 (87.6)	
Mean time in days (SD) to ART initiation among patients initiating ART within 12 months of enrollment	33.5 (64.3)	42.3 (71.2)	15.7 (42.0)	< 0.001 ^c
Median time in days (IQR) to ART initiation among patients initiating ART within 12 months of enrollment	7 (0, 32)	14 (0, 44)	0 (0, 14)	< 0.001 ^b
Clinic characteristics				
Location				
Urban/mostly urban	52,737 (78.8)	36,467 (78.5)	16,270 (79.4)	0.004
Rural/mostly rural	14,226 (21.2)	10,017 (21.5)	4,209 (20.6)	
Facility type			· · · · · · · · · · · · · · · · · · ·	
Health center	16,321 (24.4)	11,152 (24.0)	5,169 (25.2)	< 0.001
District hospital	16,559 (24.7)	11,618 (25.0)	4,941 (24.1)	
Regional, provincial or university hospital	30,301 (45.3)	20,941 (45.0)	9,360 (45.7)	
Other ^d	3,782 (5.6)	2,773 (6.0)	1,009 (4.9)	
Country income level (2018)				
Low income	24,469 (36.5)	16,600 (35.7)	7,869 (38.4)	< 0.001
Lower-middle income	31,277 (46.7)	21,887 (47.1)	9,390 (45.9)	
Upper-middle income	2,419 (3.6)	1,725 (3.7)	694 (3.4)	
High income	8,798 (13.1)	6,272 (13.5)	2,526 (12.3)	
Geographic region				
Asia-Pacific	952 (1.4%)	580 (1.2%)	372 (1.8%)	< 0.001
Central/South America	9,421 (14.1%)	6,335 (13.6%)	3,086 (15.1%)	
Central Africa	7,019 (10.5%)	4,946 (10.6%)	2,073 (10.1%)	
East Africa	42,093 (62.9%)	29,354 (63.1%)	12,739 (62.2%)	
North America	7,478 (11.2%)	5,269 (11.3%)	2,209 (10.8%)	
Year of national adoption of universal HIV treatment guidelines				

Table 1. Baseline characteristics among patients with at least 12 months of potential follow-up time after enrollment, by period of enrollment (before vs. after adoption of universal HIV treatment guidelines).

(Continued)

Table 1. (Continued)

Patient characteristics	N (%)	Before guideline adoption	After guideline adoption	<i>p</i> -Value ^a
2012–2015 ^e	8,633 (12.9)	6,091 (13.1)	2,542 (12.4)	< 0.001
2016 ^f	50,471 (75.4)	34,705 (74.7)	15,766 (77.0)	
2017–2018 ^g	7,859 (11.7)	5,688 (12.2)	2,171 (10.6)	

ART, antiretroviral therapy; IQR, interquartile range; SD, standard deviation.

^aChi-squared test.

^bKruskal–Wallis test.

^ct Test (Satterthwaite).

^dOther facility types are sites that report data as part of a network comprising clinics and hospitals.

^eArgentina, Australia, Brazil, Canada, South Korea, Mexico, Thailand, United States.

^fBurundi, Cambodia, Cameroon, Haiti, Hong Kong, Japan, Kenya, Rwanda, Uganda.

^gChile, Congo, Democratic Republic of Congo, Honduras, Malaysia, Peru, Tanzania, Vietnam.

https://doi.org/10.1371/journal.pmed.1004367.t001

with 12 months of follow-up time, with few appreciable differences between patients enrolling after versus before adoption of universal treatment guidelines (S1 Table).

Among all patients with at least 12 months of follow-up time after enrollment, 69.2% were retained in care at 12 months after enrollment, 24.2% were LTC, 3.3% were documented as deceased, and 3.4% were documented transfers (Table 2). Compared with patients enrolling before adoption of universal treatment guidelines, higher proportions of patients enrolling after guideline adoption were recorded as transfers at 12 months (4.1% versus 3.0%; p < 0.001), 24 months (6.0% versus 4.5%; p < 0.001), and 36 months after enrollment (6.7% versus 5.6%; p < 0.001) or were LTC at each time point (12 months: 25.1% versus 23.8%, p < 0.001; 24 months: 35.9% versus 31.0%, p < 0.001; 36 months: 40.2% versus 37.2%, p < 0.001). Among all patients with a record of ART initiation, 73.6% were retained in care at 12 months after ART initiation, with 62.0% and 53.4%, respectively, retained at 24 and 36 months after ART initiation. Among patients on ART, clinic retention was higher at each time point among those enrolling in care before guideline adoption than after (75.1% versus 70.5% at 12 months, p < 0.001; 64.8% versus 55.8% at 24 months, p < 0.001; and 55.0% versus 48.2% at 36 months, p < 0.001).

Among all patients retained in care after ART initiation, VL monitoring ranged from 66.7% at 12 months after ART initiation to 71.1% at 24 months and 70.0% at 36 months (Table 2), with higher proportions of patients enrolling after adoption of universal treatment guidelines having VL monitoring at 12 and 24 months (73.2% and 73.4%, respectively), compared with before (63.8% and 70.3%). In contrast, VL monitoring at 36 months was lower among patients enrolling after guideline adoption (63.7%) than before (71.6%, p < 0.001). Among patients retained after ART initiation with VL test results, VS ranged from 86.3% at 12 months to 89.1% at 36 months, with marginal increases in VS at 24 and 36 months among patients enrolling after adoption of universal HIV treatment guidelines (88.8% and 90.3%, respectively) than before (87.2% and 88.8%, respectively).

The crude cumulative incidence of LTC (CI-LTC) at each time point is shown in Fig 4, stratified by timing of enrollment relative to national adoption of universal treatment guidelines. At each time point, CI-LTC was higher among those enrolling after guideline adoption than before, with insubstantial differences at 12 months after enrollment (CI-LTC: 25.1% versus 23.8%), and larger differences at 24 months (CI-LTC: 35.9% versus 31.0%) and 36 months

Table 2.	HIV care outcomes by	v 12, 24, and 36 months aft	er enrollment, overall and by	ART initiation status by	censoring time point.
----------	----------------------	-----------------------------	-------------------------------	--------------------------	-----------------------

Care outcome	Cohort (N)	Among all patients in cohort	Before guideline adoption	After guideline adoption	<i>p</i> -Value ^a
		n (%)	n (%)	n (%)	
Death					
12 months after enrollment	66,963	2,185 (3.3)	1,503 (3.2%)	682 (3.3%)	0.516
24 months after enrollment	64,110	2,623 (4.1)	1,817 (4.0%)	806 (4.2%)	0.219
36 months after enrollment	54,251	2,471 (4.6)	1,978 (4.6%)	493 (4.3%)	0.191
Transfer					
12 months after enrollment	66,963	2,249 (3.4)	1,404 (3.0%)	845 (4.1%)	< 0.001
24 months after enrollment	64,110	3,177 (5.0)	2,045 (4.5%)	1,132 (6.0%)	< 0.001
36 months after enrollment	54,251	3,147 (5.8)	2,380 (5.6%)	767 (6.7%)	< 0.001
LTC					
12 months after enrollment	66,963	16,191 (24.2)	11,056 (23.8%)	5,135 (25.1%)	< 0.001
24 months after enrollment	64,110	20,820 (32.5)	13,997 (31.0%)	6,823 (35.9%)	< 0.001
36 months after enrollment	54,251	20,526 (37.8)	15,946 (37.2%)	4,580 (40.2%)	< 0.001
Retention in care					
12 months after enrollment	66,963	46,338 (69.2)	32,521 (70.0%)	13,817 (67.5%)	< 0.001
24 months after enrollment	64,110	37,490 (58.5)	27,239 (60.4%)	10,251 (53.9%)	< 0.001
36 months after enrollment	54,251	28,107 (51.8)	22,555 (52.6%)	5,552 (48.7%)	< 0.001
Retention in care					
12 months after ART initiation	57,615	42,418 (73.6)	29,483 (75.1%)	12,935 (70.5%)	< 0.001
24 months after ART initiation	55,416	34,350 (62.0)	24,785 (64.8%)	9,565 (55.8%)	< 0.001
36 months after ART initiation	46,850	25,041 (53.4)	20,025 (55.0%)	5,016 (48.2%)	< 0.001
VL testing among patients initiating ART and retained in care					
12 months after ART initiation	42,418	28,289 (66.7)	18,804 (63.8%)	9,485 (73.3%)	< 0.001
24 months after ART initiation	34,350	24,430 (71.1)	17,413 (70.3%)	7,017 (73.4%)	< 0.001
36 months after ART initiation	25,041	17,523 (70.0)	14,328 (71.6%)	3,195 (63.7%)	< 0.001
VS among retained patients with VL test					
12 months after ART initiation	28,289	24,404 (86.3)	16,210 (86.2%)	8,194 (86.4%)	0.671
24 months after ART initiation	24,430	21,413 (87.7)	15,184 (87.2%)	6,229 (88.8%)	0.001
36 months after ART initiation	17,523	15,606 (89.1)	12,721 (88.8%)	2,885 (90.3%)	0.013

ART, antiretroviral therapy; LTC, lost to clinic; VL, viral load; VS, viral suppression. ^aChi-squared test.

https://doi.org/10.1371/journal.pmed.1004367.t002

(CI-LTC: 40.2% versus 37.2%). CI-LTC stratified by ART initiation status is shown in Fig 5. Among patients enrolling before adoption of universal treatment guidelines, CI-LTC at each time point was more than twice as high among patients not on ART, compared with those initiated on ART, with smaller differences by ART initiation status among those enrolling after guideline adoption. Among patients already on ART, CI-LTC at each time point was substantially higher among those enrolling after the adoption of universal treatment guidelines than before; among those not yet on ART, CI-LTC at 24 and 36 months did not differ by period of enrollment (before versus after guideline adoption).

Crude and adjusted hazard ratios (aHRs) of LTC at each time point after enrollment are shown in Table 3, for all patients and stratified by ART initiation status. In the full sample, hazards of LTC at each time point were higher among patients enrolling after adoption of universal treatment guidelines compared with before, but differences were small and not statistically significant. However, the association of universal treatment guidelines with LTC hazards at 24



Fig 4. Crude CI (95% CIs) of LTC at 12, 24, and 36 months after enrollment before vs. after adoption of universal HIV treatment guidelines. ART, antiretroviral therapy; CI, cumulative incidence; LTC, lost to clinic; 95% CI, 95% confidence interval.

https://doi.org/10.1371/journal.pmed.1004367.g004

and 36 months after enrollment differed by ART initiation status. Among patients on ART by 24 and 36 months after enrollment, hazards of LTC were substantially higher among patient enrolling after adoption of universal treatment guidelines (24-month aHR 1.38 [95% CI 1.19, 1.59]; p < 0.001 and 36-month aHR 1.34 [95% CI: 1.18, 1.53]; p < 0.001). In contrast, among



Fig 5. Crude CI (95% CIs) of LTC at 12, 24, and 36 months after enrollment before vs. after adoption of universal HIV treatment guidelines, by treatment initiation status (on ART vs. not on ART) at ascertainment time point. ART, antiretroviral therapy; CI, cumulative incidence; LTC, lost to clinic; 95% CI, 95% confidence interval.

https://doi.org/10.1371/journal.pmed.1004367.g005

Care outcome	Enrollment before guideline adoption* N (%)	Enrollment after guideline adoption N (%)	HR (95% CI)	aHR ^{†§} (95% CI; <i>p</i> -value)
LTC				
12 months after enrollment [♠]	11,056 (23.8)	5,135 (25.1)	1.08 (0.96, 1.21)	1.04 (0.94, 1.15); p = 0.454
24 months after enrollment [‡]	13,997 (31.0)	6,823 (35.9)	1.19 (1.05, 1.36)	1.13 (0.99, 1.27); p = 0.056
36 months after enrollment**	15,946 (37.2)	4,580 (40.2)	1.11 (0.98, 1.27)	1.11 (0.98, 1.25); p = 0.100
LTC among patients on ART before end of follow-up				
12 months after enrollment	6,271 (17.2)	3,825 (21.3)	1.30 (1.12, 1.51)	1.25 (1.08, 1.44); p = 0.003
24 months after enrollment	9,038 (24.2)	5,510 (32.4)	1.44 (1.25, 1.67)	1.38 (1.19, 1.59); p < 0.001
36 months after enrollment	11,073 (30.7)	3,823 (36.9)	1.30 (1.13, 1.50)	1.34 (1.18, 1.53); p < 0.001
LTC among patients not on ART before end of follow-up				
12 months after enrollment	4,785 (47.7)	1,310 (51.7)	1.17 (1.00, 1.36)	1.16 (0.96, 1.39); p = 0.117
24 months after enrollment	4,959 (63.8)	1,313 (64.7)	1.03 (0.88, 1.19)	0.99 (0.86, 1.15); p = 0.934
36 months after enrollment	4,873 (72.1)	757 (72.6)	1.01 (0.87, 1.16)	1.03 (0.90, 1.18); p = 0.656

Table 3. Risks and hazard ratios of LTC associated with national adoption of universal HIV treatment guidelines.

aHR, adjusted hazards ratio; ART, antiretroviral therapy; HR, hazards ratio; LTC, lost to clinic.

*Reference group: Patients enrolling in care before adoption of universal HIV treatment guidelines.

[†]Adjusted for sex, age group, enrollment CD4, facility type, clinic location, and country income level.

[§]Transfer and death treated as competing events.

• *p*-Value for interaction term between enrollment period and ART status by censoring time point: 0.400.

 $^{*}p$ -Value for interaction term between enrollment period and ART status by censoring time point: 0.002.

**p-Value for interaction term between enrollment period and ART status by censoring time point: 0.027.

https://doi.org/10.1371/journal.pmed.1004367.t003

patients not on ART by 12, 24, and 36 months after enrollment, the hazards of LTC did not differ by period of enrollment.

Table 4 presents the proportion of patients retained in care at clinic of enrollment at 12, 24, and 36 months after ART initiation, having a record of VL monitoring and having VS at each time point, along with crude and adjusted relative risks (aRRs) of each outcome by period of enrollment. Adjusting for patient and clinic characteristics, those enrolling in care after adoption of universal treatment guidelines were less likely to be retained in care at 12 months (aRR 0.95 [95% CI: 0.93, 0.98]; p = 0.001); 24 months (aRR 0.88 [95% CI: 0.84, 0.94]; p < 0.001), and 36 months after ART initiation (aRR 0.87 [95% CI: 0.82, 0.92]; p < 0.001). Among patients retained after ART initiation, those enrolling in HIV care after adoption of universal treatment guidelines were more likely to have VL monitoring at 12 months after ART initiation (aRR 1.15 [95% CI: 1.05, 1.26]; p = 0.004) and less likely at 36 months (aRR 0.86 [95% CI: 0.80, 0.92]; p < 0.001), with no difference at 24 months. Among patients with VL monitoring at 12, 24, and 36 months after ART initiation, the likelihood of VS at 24 months was marginally, but not substantively, higher among patients enrolling in HIV care after guideline adoption (aRR 1.03 [95% CI: 1.01, 1.04]; p = 0.001), with no differences at 12 or 36 months.

Care outcome	Enrollment before guideline adoption* N (%)	Enrollment after guideline adoption N (%)	RR (95% CI)	aRR [†] (95% CI); <i>p</i> -value
Retention in care				
12 months after ART initiation	29,483 (75.1)	12,935 (70.5)	0.94 (0.91, 0.97)	0.95 (0.93, 0.98); p = 0.001
24 months after ART initiation	24,785 (64.8)	9,565 (55.8)	0.86 (0.81, 0.91)	0.88 (0.84, 0.94); p < 0.001
36 months after ART initiation	20,025 (55.0)	5,016 (48.2)	0.88 (0.82, 0.93)	0.87 (0.82, 0.92); p < 0.001
VL testing among patients initiating ART and retained in care				
12 months after ART initiation	18,804 (63.8)	9,485 (73.3)	1.15 (1.04, 1.28)	1.15 (1.05, 1.26); p = 0.004
24 months after ART initiation	17,413 (70.3)	7,017 (73.4)	1.04 (0.99, 1.10)	1.03 (0.99, 1.07); p = 0.166
36 months after ART initiation	14,328 (71.6)	3,195 (63.7)	0.89 (0.80, 0.99)	0.86 (0.80, 0.92); p < 0.001
VS among those retained in care with VL testing				
12 months after ART initiation	14,106 (86.0)	7,050 (86.2)	1.00 (0.98, 1.03)	1.01 (0.98, 1.04); p = 0.360
24 months after ART initiation	15,184 (87.2)	6,229 (88.8)	1.02 (1.00, 1.03)	1.03 (1.01, 1.04); p = 0.001
36 months after ART initiation	12,721 (88.8)	2,885 (90.3)	1.02 (1.00, 1.03)	1.01 (0.99, 1.03); p = 0.363

Table 4. Risks of HIV care outcomes after ART initiation associated with national adoption of universal HIV treatment guidelines.

aRR, adjusted risk ratio; ART, antiretroviral therapy; RR, risk ratio; VL, viral load; VS, viral suppression; 95% CI, 95% confidence interval.

*Reference group: Patients enrolling in care before adoption of universal treatment guidelines.

[†]Adjusted for sex, age group, enrollment CD4, initial regimen type, clinic location, facility type, and country income level.

https://doi.org/10.1371/journal.pmed.1004367.t004

Sensitivity analyses restricted to patients enrolling in the 13 to 24 months before adoption of universal HIV treatment guidelines and the first year afterwards showed consistent results for LTC outcomes at 12 months after enrollment, as well as retention, VL monitoring, and VS at 12 months after ART initiation (S2 Table). Post hoc sensitivity analyses that excluded countries with late adoption of universal HIV treatment guidelines in 2017 and 2018, where 24- and 36-month outcomes windows could have partially coincided with service disruptions related to the COVID-19 pandemic, also yielded results consistent with our main analyses (S3 and S4 Tables).

Discussion

Using real-world service delivery data from 109 clinics across 25 countries, our study found that patient retention at 12, 24, and 36 months after ART initiation decreased among patients enrolling in care after national adoption of universal HIV treatment guidelines. Additionally, while VS was high before and after guideline adoption among patients retained in care and on ART, there was little improvement in annual VL monitoring.

Prior experimental and observational studies examining HIV care outcomes under universal treatment guidelines have focused on outcomes within 12 months after treatment initiation [9–13,15,16] and have reported mixed results in terms of patient attrition and retention. To our knowledge, no other studies using real-world service delivery data have examined longerterm HIV care outcomes at 24 and 36 months after enrollment and after ART initiation in the era of universal HIV treatment.

In accordance with prior research [4,5,11], we found that patients enrolling in care under universal HIV treatment guidelines initiated ART more rapidly, with a lower proportion of patients having no record of ART initiation. While observed improvements in ART initiation are encouraging, our study raises concerns about the capacity of HIV programs to support engagement in care for some patients after ART initiation. Study findings also raise concerns about the quality of HIV care in the era of universal HIV treatment. WHO has recommended annual VL monitoring after ART initiation since 2013 [24]; however, VL monitoring remains suboptimal, with more than one-quarter of patients at each annual time point in our study having no record of VL testing. Among patients enrolling after national adoption of universal treatment guidelines, we found no improvement in VL testing at 24 months, and VL monitoring at 36 months decreased among patients enrolling after guideline adoption. Observed decreases in VL monitoring at 36 months are surprising because the subset of patients enrolling in HIV care after guideline adoption with at least 36 months of follow-up time comprised more patients from high-income countries with greater capacity for VL testing [25]. Additionally, while the COVID-19 pandemic is known to have disrupted HIV-related services in many settings [26,27], most patients in our study (>95%) were in countries adopting universal treatment guidelines in 2016 or earlier, meaning that VL monitoring at 36 months after ART initiation reflected prepandemic VL monitoring practices. In sensitivity analyses that excluded countries where outcomes at 36 months could have coincided with the COVID-19 pandemic findings were consistent with our main analyses. Accordingly, observed gaps in annual VL monitoring among retained patients-particularly with increasing time since ART initiationreinforce concerns about the adequacy of resource allocation for elements of HIV care that are essential for identifying adherence challenges and drug resistance and for guiding timely regimen switching [28]. Additionally, as other research has shown that patient monitoring is positively associated with retention [29], gaps in VL monitoring may represent missed opportunities for motivating patients to remain engaged in care.

Study limitations include well-known limitations of observational studies for causal attribution, along with possible underascertainment of deaths and transfers to other sites of care in routine HIV service delivery data [30–32]. As tracing studies have reported rates of undocumented (i.e., "silent") transfer ranging from 4% to 54% among patients lost to follow-up [30– 32], true LTC may be overestimated in our study, and deaths and transfers are likely underestimated [33]. Although WHO began recommending the decentralization of HIV care in LMICs in the mid- to late 2000s, well before the adoption of universal HIV treatment guidelines in these settings [34–36] and we used a conservative definition of LTC [37], it is possible that decentralization has accelerated with the rollout of universal treatment policies, resulting more undocumented transfers, particularly among patients at centralized or tertiary care sites who silently transfer to peripheral HIV care facilities [31,32].

While we were able to adjust for selected patient characteristics, there may be important unmeasured differences between patients enrolling in HIV care and initiating treatment before and after the adoption of universal treatment guidelines. Other analyses have shown that patients entering HIV care in the era of universal treatment initiate treatment more rapidly [4,5] and that the expansion of HIV treatment eligibility criteria has resulted in the treatment of patients with earlier stage HIV disease [38]. While earlier treatment initiation is associated with improved clinical outcomes and reduced onward HIV transmission, rapid initiation of treatment after enrollment in HIV care is also associated with lower retention in care [39,40], and qualitative research has suggested that distress and uncertainty about HIV diagnosis, concerns about stigma, fear of lifelong medication, and other patient-level factors may contribute to attrition and lower treatment adherence among those rapidly initiating treatment, particularly among patients with early stage disease who do not feel unwell [41,42]. Our observed

decreases in retention after ART initiation in the era of universal treatment may reflect the fact that some patients, who would have been LTC prior to ART initiation before guideline adoption, initiated ART more rapidly.

We had limited data on patient characteristics and note that substantial decreases in CD4 testing after the adoption of universal treatment guidelines [43] constrain our ability to adjust for patient immunological status at the time of care entry and treatment initiation, which may be a source of bias in the estimates we report. We also were unable to adjust for pregnancy status or examine whether longer-term HIV care outcomes among pregnant women-a group that was eligible for immediate treatment and life-long ART prior to the adoption of universal HIV treatment guidelines-differed from those of other patients. Additionally, while we adjusted for clinic characteristics, such as location, facility type, and country income level, there may be important time-varying contextual and health system factors, including supplyside constraints related to increased demand for HIV treatment, which we were unable to adjust for in our analyses. It is noteworthy that for patients in our study who enrolled in HIV care and initiated ART before the adoption of universal HIV treatment guidelines, outcomes at 24 and 36 months occurred after guidelines had changed. While it is unlikely that routine follow-up care of patients established on ART differed by the timing of treatment initiation, practices related to ART readiness and adherence counseling may have differed substantially before and after the adoption of universal HIV treatment guidelines but we were unable to examine or control for such differences.

A further limitation is the nonrepresentativeness of IeDEA sites within countries and regions included in this analysis; almost half of our sample were in care at university and tertiary referral hospitals, and these sites are likely better resourced than other HIV clinics within the same geographic areas. Accordingly, our estimates of VL monitoring before and after the adoption of universal HIV treatment guidelines may be biased upwards. While consistent with other research [44,45], our estimates of VS may also be biased upwards, particularly as those without VL monitoring likely include patients disengaged from care—patients known to have higher rates of viral nonsuppression [46,47]. Additionally, our study findings may not be generalizable to other locations and contexts that were not included in our study. It is also possible that patients enrolling in care and initiating treatment during the first year after the adoption of universal HIV treatment guidelines are not representative of those enrolling during subsequent years, with HIV outcomes improving among patients enrolling 2 to 3 years after the initial period of guideline adoption.

Key strengths of our study include the use of real-world service delivery data from a large sample to examine longer-term programmatic outcomes after the adoption of universal HIV treatment guidelines across diverse country settings. Associations observed in our heterogeneous sample of patients from diverse settings across multiple regions and different years of guideline adoption may be broadly reflective of HIV care outcomes among patients who were ART naïve at enrollment in HIV care during the years surrounding the adoption of universal HIV treatment guidelines in these settings.

While the adoption of universal HIV treatment guidelines has expanded access to and uptake of timely treatment for PLWH, our findings raise concerns about existing service delivery strategies and capacity for longer-term retention of patients in the era of universal treatment. Our findings of increased LTC at 24 and 36 months after enrollment and decreased retention at all time points after ART initiation suggest a risk of worsened patient outcomes. Additionally, while VS rates at all time points are high, our study highlights suboptimal VL monitoring, particularly with increased duration of time in care. Because of lags in the availability of data extracted from patient records and databases, we were only able to examine outcomes among patients enrolling in HIV care and initiating treatment during the first year after

the adoption of universal treatment guidelines. However, our findings underscore the critical importance of monitoring long-term HIV care outcomes as additional data become available, as well as examining HIV care outcomes among groups known to be at increased risk of attrition and poor VS, including pediatric patients and pregnant and postpartum women. Equally vital are efforts to identify and address health system, community and patient-level determinants of attrition before and after ART initiation, and barriers to adherence and viral nonsuppression in the era of universal treatment of all PLWH.

Supporting information

S1 Checklist. STROBE Statement. (DOCX)

S1 Text. Acknowledgments. (DOCX)

S1 Concept Proposal. Concept sheet: Multiregional analysis. (PDF)

S1 Table. Baseline characteristics among patients enrolling in care at least 24 and 36 months before database closure. (DOCX)

S2 Table. Sensitivity analyses restricted to patients enrolling 13–24 months before and 12 months after national adoption of universal HIV treatment guidelines. (DOCX)

S3 Table. Risks and hazards of LTC associated with national adoption of universal HIV treatment guidelines in countries introducing guideline changes before 2017. (DOCX)

S4 Table. Relative risks of HIV care outcomes after ART initiation associated with national adoption of universal HIV treatment guidelines in countries introducing guideline changes before 2017.

(DOCX)

Acknowledgments

We sincerely thank the staff at contributing sites, as well as IeDEA regional data managers. The full list of Acknowledgments can be found in the Supporting information (S1 Text).

Author Contributions

Conceptualization: Olga Tymejczyk, Denis Nash.

Data curation: Kara Wools-Kaloustian, Awachana Jiamsakul, Marco Tulio Luque Torres, Jennifer S. Lee, Lisa Abuogi, Vohith Khol, Fernando Mejía Cordero, Keri N. Althoff, Matthew G. Law.

Formal analysis: Ellen Brazier.

Funding acquisition: Kara Wools-Kaloustian, Keri N. Althoff, Matthew G. Law, Denis Nash.

Methodology: Ellen Brazier, Olga Tymejczyk, Denis Nash.

Writing – original draft: Ellen Brazier.

Writing – review & editing: Ellen Brazier, Olga Tymejczyk, Kara Wools-Kaloustian, Awachana Jiamsakul, Marco Tulio Luque Torres, Jennifer S. Lee, Lisa Abuogi, Vohith Khol, Fernando Mejía Cordero, Keri N. Althoff, Matthew G. Law, Denis Nash.

References

- World Health Organization. Guideline on when to start antiretroviral therapy and on pre-exposure prophylaxis for HIV. Geneva, Switzerland: World Health Organization; 2015 [accessed 2022 Oct 17]. Available from: https://www.who.int/publications/i/item/9789241509565.
- 2. Nash D, Yotebieng M, Sohn AH. Treating all people living with HIV in sub-Saharan Africa: a new era calling for new approaches. J Virus Erad. 2018; 4(Suppl 2):1–4. PMID: 30515307
- Ford N, Vitoria M, Doherty M. Providing antiretroviral therapy to all who are HIV positive: the clinical, public health and programmatic benefits of Treat All. J Int AIDS Soc. 2018; 21(2):e25078. <u>https://doi.org/10.1002/jia2.25078</u> PMID: 29436776
- Tymejczyk O, Brazier E, Yiannoutsos CT, Vinikoor M, van Lettow M, Nalugoda F, et al. Changes in rapid HIV treatment initiation after national "treat all" policy adoption in 6 sub-Saharan African countries: Regression discontinuity analysis. PLoS Med. 2019; 16(6):e1002822. https://doi.org/10.1371/journal. pmed.1002822 PMID: 31181056
- Tymejczyk O, Brazier E, Wools-Kaloustian K, Davies MA, Dilorenzo M, Edmonds A, et al. Impact of Universal Antiretroviral Treatment Eligibility on Rapid Treatment Initiation Among Young Adolescents with Human Immunodeficiency Virus in Sub-Saharan Africa. J Infect Dis. 2020; 222(5):755–764. <u>https://doi.org/10.1093/infdis/jiz547</u> PMID: 31682261
- O'Brien WA, Hartigan PM, Martin D, Esinhart J, Hill A, Benoit S, et al. Changes in plasma HIV-1 RNA and CD4+ lymphocyte counts and the risk of progression to AIDS. Veterans Affairs Cooperative Study Group on AIDS. N Engl J Med. 1996; 334(7):426–431. <u>https://doi.org/10.1056/</u> NEJM199602153340703 PMID: 8552144
- Joint United Nations Programme on HIV/AIDS. Public Health and HIV Viral Load Suppression. Geneva: UNAIDS; 2017 [accessed 2018 Mar 15]. Available from: http://www.unaids.org/sites/default/files/ media_asset/20170724_viral_load_suppression_brochure.pdf.
- Cohen MS, Chen YQ, McCauley M, Gamble T, Hosseinipour MC, Kumarasamy N, et al. Prevention of HIV-1 infection with early antiretroviral therapy. N Engl J Med. 2012; 365(6):493–505.
- Onoya D, Hendrickson C, Sineke T, Maskew M, Long L, Bor J, et al. Attrition in HIV care following HIV diagnosis: a comparison of the pre-UTT and UTT eras in South Africa. J Int AIDS Soc. 2021; 24(2): e25652. https://doi.org/10.1002/jia2.25652 PMID: 33605061
- Chauke P, Huma M, Madiba S. Lost to follow up rate in the first year of ART in adults initiated in a universal test and treat programme: a retrospective cohort study in Ekurhuleni District, South Africa. Pan Afr Med J. 2020; 37:198. https://doi.org/10.11604/pamj.2020.37.198.25294 PMID: 33505567
- Mody A, Sikazwe I, Namwase AS, Wa Mwanza M, Savory T, Mwila A, et al. Effects of implementing universal and rapid HIV treatment on initiation of antiretroviral therapy and retention in care in Zambia: a natural experiment using regression discontinuity. Lancet HIV. 2021; 8(12):e755–e765. https://doi.org/ 10.1016/S2352-3018(21)00186-7 PMID: 34656208
- Hirasen K, Fox MP, Hendrickson CJ, Sineke T, Onoya D. HIV Treatment Outcomes Among Patients Initiated on Antiretroviral Therapy Pre and Post-Universal Test and Treat Guidelines in South Africa. Ther Clin Risk Manag. 2020; 16:169–180. https://doi.org/10.2147/TCRM.S227290 PMID: 32184609
- Alhaj M, Amberbir A, Singogo E, Banda V, van Lettow M, Matengeni A, et al. Retention on antiretroviral therapy during Universal Test and Treat implementation in Zomba district, Malawi: a retrospective cohort study. J Int AIDS Soc. 2019; 22(2):e25239. https://doi.org/10.1002/jia2.25239 PMID: 30734510
- Filiatreau LM, Edwards JK, Masilela N, Gómez-Olivé FX, Haberland N, Pence BW, et al. Understanding the effects of universal test and treat on longitudinal HIV care outcomes among South African youth: a retrospective cohort study. BMC Public Health. 2023; 23(1):1724. <u>https://doi.org/10.1186/s12889-023-16353-9 PMID: 37670262</u>
- 15. Khan S, Spiegelman D, Walsh F, Mazibuko S, Pasipamire M, Chai B, et al. Early access to antiretroviral therapy versus standard of care among HIV-positive participants in Eswatini in the public health sector: the MaxART stepped-wedge randomized controlled trial. J Int AIDS Soc. 2020; 23(9):e25610. https://doi.org/10.1002/jia2.25610 PMID: 32949103
- Brown LB, Havlir DV, Ayieko J, Mwangwa F, Owaraganise A, Kwarisiima D, et al. High levels of retention in care with streamlined care and universal test and treat in East Africa. AIDS. 2016; 30(18):2855–2864. https://doi.org/10.1097/QAD.0000000001250 PMID: 27603290

- IeDEA. International Epidemiologic Databases to Evaluate AIDS. Bern, Switzerland: the Institute of Social and Preventive Medicine (ISPM), University of Bern; [accessed 2023 Dec 1]. Available from: https://www.iedea.org/.
- Duda SN, Musick BS, Davies M-A, Sohn AH, Ledergerber B, Wools-Kaloustian K, et al. The IeDEA Data Exchange Standard: a common data model for global HIV cohort collaboration. medRxiv [Preprint]. 2020:2020.07.22.20159921.
- Brazier E, Maruri F, Duda SN, Tymejczyk O, Wester CW, Somi G, et al. Implementation of "Treat-all" at adult HIV care and treatment sites in the Global IeDEA Consortium: results from the Site Assessment Survey. J Int AIDS Soc. 2019; 22(7):e25331. https://doi.org/10.1002/jia2.25331 PMID: 31623428
- Edwards JK, Cole SR, Westreich D, Moore R, Mathews C, Geng E, et al. Loss to clinic and five-year mortality among HIV-infected antiretroviral therapy initiators. PLoS ONE. 2014; 9(7):e102305. <u>https:// doi.org/10.1371/journal.pone.0102305</u> PMID: 25010739
- The World Bank. World Bank list of economies, June 2017. 2017 [accessed 2017 Nov 15]. Available from: https://datahelpdesk.worldbank.org/knowledgebase/articles/906519-world-bank-country-andlending-groups.
- Aalen OO, Johansen S. An Empirical Transition Matrix for Non-Homogeneous Markov Chains Based on Censored Observations. Scand J Stat. 1978; 5(3):141–150.
- Andersen PK, Geskus RB, de Witte T, Putter H. Competing risks in epidemiology: possibilities and pitfalls. Int J Epidemiol. 2012; 41(3):861–870. https://doi.org/10.1093/ije/dyr213 PMID: 22253319
- 24. World Health Organization (WHO). Consolidated guidelines on the use of antiretroviral drugs for treating and preventing HIV infection: recommendations for a public health approach. Geneva: World Health Organization; 2013 [accessed 2019 Jan 31]. Available from: https://apps.who.int/iris/bitstream/ handle/10665/85321/9789241505727_eng.pdf?sequence=1.
- 25. Pham MD, Nguyen HV, Anderson D, Crowe S, Luchters S. Viral load monitoring for people living with HIV in the era of test and treat: progress made and challenges ahead—a systematic review. BMC Public Health. 2022; 22(1):1203. https://doi.org/10.1186/s12889-022-13504-2 PMID: 35710413
- 26. Brazier E, Ajeh R, Maruri F, Musick B, Freeman A, Wester CW, et al. Service delivery challenges in HIV care during the first year of the COVID-19 pandemic: results from a site assessment survey across the global IeDEA consortium. J Int AIDS Soc. 2022; 25(12):e26036. <u>https://doi.org/10.1002/jia2.26036</u> PMID: 36504431
- 27. Rogers A, Brazier E, Dzudie A, Adedimeji A, Yotebieng M, Muhoza B, et al. COVID-19 associated changes in HIV service delivery over time in Central Africa: Results from facility surveys during the first and second waves of the pandemic. PLoS ONE. 2022; 17(11):e0275429. <u>https://doi.org/10.1371/journal.pone.0275429 PMID: 36449454</u>
- Peter T, Ellenberger D, Kim AA, Boeras D, Messele T, Roberts T, et al. Early antiretroviral therapy initiation: access and equity of viral load testing for HIV treatment monitoring. Lancet Infect Dis. 2017; 17(1): e26–e29. https://doi.org/10.1016/S1473-3099(16)30212-2 PMID: 27773596
- Harlow AF, Bor J, Brennan AT, Maskew M, MacLeod W, Carmona S, et al. Impact of Viral Load Monitoring on Retention and Viral Suppression: A Regression Discontinuity Analysis of South Africa's National Laboratory Cohort. Am J Epidemiol. 2020; 189(12):1492–1501. https://doi.org/10.1093/aje/kwaa140 PMID: 32648905
- 30. Ballif M, Christ B, Anderegg N, Chammartin F, Muhairwe J, Jefferys L, et al. Tracing People Living With Human Immunodeficiency Virus Who Are Lost to Follow-up at Antiretroviral Therapy Programs in Southern Africa: A Sampling-Based Cohort Study in 6 Countries. Clin Infect Dis. 2022; 74(2):171–179.
- Wilkinson LS, Skordis-Worrall J, Ajose O, Ford N. Self-transfer and mortality amongst adults lost to follow-up in ART programmes in low- and middle-income countries: systematic review and meta-analysis. Trop Med Int Health. 2015; 20(3):365–379. https://doi.org/10.1111/tmi.12434 PMID: 25418366
- 32. Geng EH, Odeny TA, Lyamuya R, Nakiwogga-Muwanga A, Diero L, Bwana M, et al. Retention in Care and Patient-Reported Reasons for Undocumented Transfer or Stopping Care Among HIV-Infected Patients on Antiretroviral Therapy in Eastern Africa: Application of a Sampling-Based Approach. Clin Infect Dis. 2016; 62(7):935–944. https://doi.org/10.1093/cid/civ1004 PMID: 26679625
- Jiamsakul A, Gani Y, Avihingsanon A, Azwa I, Chaiwarith R, Khusuwan S, et al. Brief Report: Mortality After Loss to Follow-Up-A Linkage Study of People Living With HIV in Thailand and Malaysia. J Acquir Immune Defic Syndr. 2022; 91(3):290–295. <u>https://doi.org/10.1097/QAI.000000000003067</u> PMID: 35969472
- Haghighat R, Steinert J, Cluver L. The effects of decentralising antiretroviral therapy care delivery on health outcomes for adolescents and young adults in low- and middle-income countries: a systematic review. Glob Health Action. 2019; 12(1):1668596. <u>https://doi.org/10.1080/16549716.2019.1668596</u> PMID: 31558145

- 35. Gilks CF, Crowley S, Ekpini R, Gove S, Perriens J, Souteyrand Y, et al. The WHO public-health approach to antiretroviral treatment against HIV in resource-limited settings. Lancet. 2006; 368 (9534):505–510. https://doi.org/10.1016/S0140-6736(06)69158-7 PMID: 16890837
- 36. Ford N, Ball A, Baggaley R, Vitoria M, Low-Beer D, Penazzato M, et al. The WHO public health approach to HIV treatment and care: looking back and looking ahead. Lancet Infect Dis. 2018; 18(3): e76–e86. https://doi.org/10.1016/S1473-3099(17)30482-6 PMID: 29066132
- Kebede HK, Mwanri L, Ward P, Gesesew HA. Predictors of lost to follow up from antiretroviral therapy among adults in sub-Saharan Africa: a systematic review and meta-analysis. Infect Dis Poverty. 2021; 10(1):33. https://doi.org/10.1186/s40249-021-00822-7 PMID: 33743815
- Tymejczyk O, Brazier E, Yiannoutsos C, Wools-Kaloustian K, Althoff K, Crabtree-Ramirez B, et al. HIV treatment eligibility expansion and timely antiretroviral treatment initiation following enrollment in HIV care: A metaregression analysis of programmatic data from 22 countries. PLoS Med. 2018; 15(3): e1002534.
- Ross J, Brazier E, Fatti G, Jaquet A, Tanon A, Haas AD, et al. Same-Day Antiretroviral Therapy Initiation as a Predictor of Loss to Follow-up and Viral Suppression Among People With Human Immunodeficiency Virus in Sub-Saharan Africa. Clin Infect Dis. 2023; 76(1):39–47. https://doi.org/10.1093/cid/ ciac759 PMID: 36097726
- Kerschberger B, Boulle A, Kuwengwa R, Ciglenecki I, Schomaker M. The Impact of Same-Day Antiretroviral Therapy Initiation Under the World Health Organization Treat-All Policy. Am J Epidemiol. 2021; 190(8):1519–1532. https://doi.org/10.1093/aje/kwab032 PMID: 33576383
- 41. Ross J, Ingabire C, Umwiza F, Gasana J, Munyaneza A, Murenzi G, et al. How early is too early? Challenges in ART initiation and engaging in HIV care under Treat All in Rwanda-A qualitative study. PLoS ONE. 2021; 16(5):e0251645. https://doi.org/10.1371/journal.pone.0251645 PMID: 33984044
- 42. Horter S, Bernays S, Thabede Z, Dlamini V, Kerschberger B, Pasipamire M, et al. "I don't want them to know": how stigma creates dilemmas for engagement with Treat-all HIV care for people living with HIV in Eswatini. Afr J AIDS Res. 2019; 18(1):27–37. <u>https://doi.org/10.2989/16085906.2018.1552163</u> PMID: 30782082
- Brazier E, Tymejczyk O, Zaniewski E, Egger M, Wools-Kaloustian K, Yiannoutsos CT, et al. Effects of National Adoption of Treat-All Guidelines on Pre-Antiretroviral Therapy (ART) CD4 Testing and Viral Load Monitoring After ART initiation: A Regression Discontinuity Analysis. Clin Infect Dis. 2021; 73(6): e1273–e1281. https://doi.org/10.1093/cid/ciab222 PMID: 33693517
- 44. Han WM, Law MG, Egger M, Wools-Kaloustian K, Moore R, McGowan C, et al. Global estimates of viral suppression in children and adolescents and adults on antiretroviral therapy adjusted for missing viral load measurements: a multiregional, retrospective cohort study in 31 countries. Lancet HIV. 2021; 8 (12):e766–e775. https://doi.org/10.1016/S2352-3018(21)00265-4 PMID: 34856180
- 45. Jiamsakul A, Kariminia A, Althoff KN, Cesar C, Cortes CP, Davies MA, et al. HIV Viral Load Suppression in Adults and Children Receiving Antiretroviral Therapy-Results From the IeDEA Collaboration. J Acquir Immune Defic Syndr. 2017; 76(3):319–329. <u>https://doi.org/10.1097/QAI.00000000001499</u> PMID: 28708808
- 46. Gopalsamy SN, Shah NS, Marconi VC, Armstrong WS, Del Rio C, Pennisi E, et al. The Impact of Churn on HIV Outcomes in a Southern United States Clinical Cohort. Open Forum Infect Dis. 2022; 9(7): ofac338. https://doi.org/10.1093/ofid/ofac338 PMID: 35899283
- Krentz HB, Vu Q, Gill MJ. The Impact of "Churn" on Plasma HIV Burden Within a Population Under Care. Open Forum Infect Dis. 2019; 6(6):ofz203. https://doi.org/10.1093/ofid/ofz203 PMID: 31205974